9448007, 2023, 5, Do



Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2022GL102184

Key Points:

- Cloud fraction (CF) of tropical cyclone (TC) high deep clouds has recently declined in the inner core while the outer-region change is much smaller
- The decline was most significant during the intensification period
- The percentage change of TC innercore CF of high thick cloud was similar to that of TC inner-core rain rate

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

T.-K. Lai, eric.lai@imperial.ac.uk

Citation:

Lai, T.-K., & Toumi, R. (2023). Has there been a recent shallowing of tropical cyclones? *Geophysical Research Letters*, 50, e2022GL102184. https://doi. org/10.1029/2022GL102184

Received 16 NOV 2022 Accepted 11 FEB 2023

© 2023. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Has There Been a Recent Shallowing of Tropical Cyclones?

Tsz-Kin Lai¹ 🝺 and Ralf Toumi¹ 🕩

¹Space and Atmospheric Physics Group, Department of Physics, Imperial College London, London, UK

Abstract Many aspects of tropical cyclone (TC) properties at the surface have been changing but any systematic vertical changes are unknown. Here, we document a recent trend of high thick clouds of TCs. The global inner-core high thick cloud fraction measured by satellite has decreased from 2002 to 2021 by about 10% per decade. The TC inner-core surface rain rate is also found to have decreased during the same period by a similar percentage. This suppression of high thick clouds and rain has been largest during the intensification phase of the strongest TCs. Hence, these two independent and consistent observations suggest that the TC inner-core convection has weakened and that TCs have become shallower recently at least. For this period, the lifetime maximum intensity of major TCs has not changed and this suggests an increased efficiency of the spin-up of TCs.

Plain Language Summary As the atmosphere is becoming warmer under the climate change, many aspects of tropical cyclone (TC) have been changing or are expected to change, for example, TC intensity and TC height. The height of stronger TCs is expected to become taller. We present the first satellite observations of TC clouds near the TC top and demonstrate that the thick clouds of the inner part of TC, which is the part with the most intense winds and rain, has actually decreased during the period between 2002 and 2021. The decrease of high thick clouds within the inner part of TC is also consistent with the observed decreased rain there. However, the lifetime maximum intensity has not changed for this period. This all suggests that the strongest convection of a TC has weakened, and become shallower. TC may also have been spinning up more efficiently converting less energy to similar maximum intensities.

1. Introduction

Tropical cyclones (TCs) may have become increasingly destructive (Emanuel, 2005; Wang & Toumi, 2022). Theory and models suggest that there will be significant increase in TC intensity (e.g., Emanuel, 1987; Hill & Lackmann, 2011; Holland, 1997; Knutson et al., 2015, 2020) in a warming climate. Most studies point to a future decrease in frequency (e.g., Knutson et al., 2020). A poleward and coastal migration of TCs has also been reported (Kossin et al., 2014; Wang & Toumi, 2021a). The TC size is correlated to TC-induced economic damages such as insured losses (Wang & Toumi, 2016; Zhai & Jiang, 2014) and its climatology has been explored (e.g., Kimball & Mulekar, 2004; Schenkel et al., 2018; Wang & Toumi, 2018) while no significant trends of TC size have been identified for the period from 1981 to 2011 (Knaff et al., 2014). Indirect methods using the sea surface temperature cold wakes (Chavas et al., 2016) also suggest no long-term change in size (Wang & Toumi, 2021b).

All of the above studies focused on the surface properties. There is a lack of evidence of any changes in the vertical characteristics. The height/depth of a TC is also an essential property (Hendricks et al., 2004; Simpson et al., 1998). Z. Wang et al. (2009) showed that pre-genesis deep convection or disturbance is conducive to genesis. The probabilities of intensification and rapid intensification increase when very tall convective towers exist in the eyewall (Jiang, 2012; Kelley et al., 2004). Schubert and McNoldy (2010) showed that stronger TC vortices have deeper secondary circulation, implying a taller convection within the eyewall. Yamada et al. (2010) demonstrated that the cloud height of simulated strong TCs will become taller under the global warming. This increase in height was associated with the warm-core structural change simulated by Knutson and Tuleya (1999). To date there have been no documented observed trends of TC cloud top height (CTH) and cloud optical depth (COD). The cloud fraction (CF) of TC deep convective cloud is of interest as a proxy for CTH and COD. It is well understood that deeper convection can yield heavier rain (e.g., Adler & Mack, 1984; Song et al., 2020). In particular the cloud depth is often inferred from infrared cloud top brightness temperatures (e.g., So & Shin, 2018); and the cloud depth is often used to estimate convective rain rate (e.g., Tapiador et al., 2019). Here we use satellite observations to show a declining trend of TC CF.

Deep convective systems and cloud cover are linked to the associated rain rate (e.g., Adler & Mack, 1984) and we will demonstrate a correlation for TCs. TC rainfall is broadly a function of TC intensity (Lonfat et al., 2004) and available moisture. The predicted increase in TC intensity combined with ocean warming projects to a robust future increase in TC rainfall (e.g., Hill & Lackmann, 2011; Knutson et al., 2015, 2020). However, despite this expectation, there have been surprising decreases of TC inner-core rainfall rate since late 1990s (Guzman & Jiang, 2021; Lavender & McBride, 2021; Tu et al., 2021). There is some uncertainty in these trends since the Tropical Rainfall Measuring Mission (TRMM) has limitations compared to better-characterized data (Huffman, 2021). Since these studies imply a shallowing/weakening of TC inner-core deep convection, we aim to answer the question "Can we use cloud observations to test for decreases in deep convection consistent with the reported decline in inner-core rainfall rate?"

2. Methods

The global best track data of TCs are obtained for US agencies in the International Best Track Archive for Climate Stewardship (IBTrACS v.4) (Knapp et al., 2010, 2018). All the non-TC systems (e.g., extra-tropical systems, tropical waves, and tropical disturbances) are excluded. 2088 TCs with wind speeds larger than 34 knots (tropical storm or stronger) between 1998 and 2021 comprise the data set. The Supporting Information S1 provides detailed information about the case selection.

The daily CTH and COD of TCs from 2002 (4 July) to 2021 are provided by Moderate Resolution Imaging Spectroradiometer (MODIS)/Aqua Level-3 Daily Cloud Properties Product (CLDPROP) version 1.1 (Platnick et al., 2019). MODIS/Aqua, which has 36 channels ranging from 0.415 to 14.235 μ m (King et al., 2003), is in sun-synchronous orbit with an equator crossing time of around 13:30 in the ascending mode. The horizontal data grid size is 1° × 1°, in which the image pixel count is sampled at a 5-km interval from the 1-km observation (Marchand et al., 2010). Deep convection is associated with intermediate ($3.6 \le OD < 23.0$) and high optical thickness ($OD \ge 23.0$; Marchand, 2013). The grid box covering the TC center is regarded as the innercore grid box which will cover $\ge 25\%$ of a typical inner core. There will be a bias in occasionally loosing some inner-core information to adjacent grids in individual cases, but this is unlikely to affect the long-term trends of large samples that we are interested in here. Outer-region grid boxes are within 500 km of the TC center (excluding the inner-core grid box). Only TC cases having at least one CLDPROP record with $\ge 25\%$ of spatial coverage during the averaging period are considered when calculating the corresponding average (lifetime, pre-lifetime maximum intensity [LMI], or post-LMI). Results are similar if we use a higher threshold value (e.g., 50%).

Three hourly TC precipitation data are extracted from the TRMM Project Multi-satellite Precipitation Analysis (TMPA) Rainfall Estimate L3 version 7 (3B42) (Huffman et al., 2007; Tropical Rainfall Measuring Mission (TRMM), 2011) with a spatial resolution of $0.25^{\circ} \times 0.25^{\circ}$. Rainfall is estimated by using microwave-only, infrared-only, and mixed microwave and infrared measurements within 500 km of the TC center at every 3-hr best-track position between 1998 and 2019 projected onto a polar coordinate centered at the TC center. We only use rain rates ≥ 0.1 mm hr⁻¹ (Guzman & Jiang, 2021). The inner core for rain is defined as the region within 2× the maximum azimuthal rain rate (Guzman & Jiang, 2021). TRMM records only within 1 hr of CLDPROP are considered for calculating the ranked correlation between the TC inner-core CF of high thick clouds and TC inner-core rain rate.

We will also examine the relationship between LMI and pre-LMI rainfall. To explore the relationship, we define *C* as the kinetic energy of *a unit mass of air* at LMI divided by the total column latent heat released in a unit area,

$$C = \frac{mV_{\text{max}}^2}{2\int_0^t \rho_{\ell w} RL \, dt},\tag{1}$$

where *m* is the mass of air, V_{max} is the maximum surface wind speed at LMI, *L* is the latent heat of vaporization, $\rho_{\ell w}$ is the liquid water density, *R* is the inner-core rain rate, and *t* is time to LMI from tropical storm intensity. The integrated heating between the condensation level and the cloud top is proportional to the surface rainfall and the effects of freezing and water vapor deposition are ignored (Yanai et al., 1973). *C* will contain some information about the efficiency of the conversion of the latent heat generated during the TC spin-up to the final kinetic energy.





Figure 1. The linear trends (per decade) of the global annual-mean cloud fraction of tropical cyclone (TC) clouds with different cloud top height within (left) the inner cores and (right) the outer regions of TCs from July 4, 2002 to December 31, 2021. The black curves are for all available TC cases while the blue curves are for CAT 3–5 TCs. The shaded areas highlight the standard errors of the linear trends.

3. Results

Figure 1 shows the linear trends of lifetime-mean CF as a function of CTH. Globally, the amount of inner-core clouds reaching $15 \le z \le 17$ km has largely decreased, while the amount of those reaching lower altitudes ($12 \le z < 15$ km) is largely increasing (Figure 1a). A similar pattern also takes place in the outer region at lower altitudes (Figure 1b). During this 20-year period, both the inner-core and outer-region CF of CTH peaks at $13 \le z \le 14$ km (~0.2; Figures S1a and S1b in Supporting Information S1). But the cumulative CF for CTH ≥ 13 km is much larger within the inner core (~0.6) than that in the outer region (~0.3), especially for major TCs (CAT 3–5). For inner-core COD, CF peaks at very thick clouds (COD ≥ 100) whilst the thick clouds (COD ≥ 23) have the highest cumulative CF (~0.6; Figure S1c in Supporting Information S1). Major TCs have clearly more thick clouds and less thin or intermediate clouds. For the outer region, CF peaks at thin clouds (COD < 3.6) and the cumulative CF of thick cloud is significantly smaller than that in the inner core.

The trend of inner-core COD largely declines among the very thick clouds while it increases among the thin and intermediate clouds (Figure S2a in Supporting Information S1). The pattern of outer-region COD is similar except that the CF of very thick cloud has much smaller rate of change (Figure S2b in Supporting Information S1). One can also see that the decline of TC CF of high deep clouds was larger among the major TCs. The observed changes in CF of inner-core clouds reaching $15 \le z \le 17$ km are -0.026 ± 0.010 (-11%) and -0.036 ± 0.016 (-12%; Figure 2a) per decade globally for all TC cases and major TCs, respectively. For the inner-core COD, the observed changes for very thick clouds are -0.020 ± 0.011 (-9%; all TCs) and -0.014 ± 0.014 (-5%; major TCs; Figure 2d).

To investigate whether the CF decline occurs during the intensification (i.e., pre-LMI) or decay (i.e., post-LMI) stages, we then further decompose the data into the pre-LMI and post-LMI periods when maximum wind speed exceeds 34 knots. For $15 \le \text{CTH} \le 17$ km, there is only a global CF decline in the pre-LMI period (Figures 2b and 2c). Similarly, for COD ≥ 100 , there is a significantly declining global pre-LMI trend of CF and no post-LMI global trend (Figures 2e and 2f). There is some variability between basins as expected (see the Supporting Information S1). But generally, lifetime decline in CTH and COD can be seen in many of the basins with pre-LMI decline much larger than their post-LMI trends.

To summarize, the CF of inner-core deep convective clouds of TCs has declined from 2002 to 2021 in the basins covering most TC activity. This suggests that the inner-core deep convection of TCs became shallower (and thus weaker) especially during the pre-LMI intensification stage.



Geophysical Research Letters

10.1029/2022GL102184



Figure 2. The time series of the global annual-mean cloud fraction of (top) tropical cyclone (TC) inner-core clouds reaching $15 \le z \le 17$ km or (bottom) TC inner-core clouds with high optical depth (COD ≥ 100) averaged over (left) the whole lifetime, (middle) the pre-lifetime maximum intensity (LMI) period, and (right) the post-LMI period from July 4, 2002 to December 31, 2021 for CAT 3–5 TCs. The shaded areas indicate the two-sided 95% confidence intervals of the linear trends with the Spearman correlation coefficient *r* and *p* value *P*.

One would expect an association between the inner-core rain rate and high thick CF but this has not yet been documented for TCs. Globally the ranked correlation is positive and very large ($r \ge 0.96$; Figures 3a and 3d). One can see that the post-LMI period has larger slopes of the linear fit (Figures 3b, 3c, 3e, and 3f). We note that the correlation coefficient is lower during the post-LMI period, especially for CTH (Figures 3c and 3f).

The combination of a strong positive correlation between the inner-core rain rate and the inner-core CF of clouds with high CTH or COD, and the decline of inner-core CF of high thick cloud naturally lead us to re-examine the reported decrease of inner-core rain rate. The inner-core rain rate has indeed declined substantially globally (Figure 4 and Figure S5 in Supporting Information S1) by 0.92 ± 0.12 mm hr⁻¹ per decade (major TCs) or 0.52 ± 0.08 mm hr⁻¹ per decade (all TCs). The outer-region rain rate increased globally only a little (not shown). The decrease in inner-core rain rate, like the CF, was larger among major TCs (comparing Figure 4 to Figure S5 in Supporting Information S1). Similar to the CF data, the rain rate is further decomposed into pre-LMI and post-LMI periods. The decrease in inner-core rain rate during the pre-LMI period is larger by 28% than the post-LMI value (e.g., Figures 4b and 4c). This is a new insight into the rain trends. This pattern of a larger rain rate decline during the pre-LMI period is similar to the pre-LMI decline of inner-core CF of cloud reaching $15 \le z \le 17$ km (e.g., Figures 2b and 2c).

Both inner-core rain rate and inner-core CF of high or thick cloud have declined in the past few decades (especially for major TCs). Table 1 shows the corresponding percentage changes over the whole globe and the basins. We can see that the percentage changes of lifetime-mean inner-core CF of cloud reaching $15 \le z \le 17$ km and inner-core rain rate are similar. This shows remarkable consistency between these independent data. The change in LMI is not significant during this time period in any basin, consistent with the relatively stable tropical SST in



Geophysical Research Letters

10.1029/2022GL102184



Figure 3. The correlation between the global mean tropical cyclone (TC) inner-core rain rate and cloud fraction (CF) of (top) inner-core clouds with $15 \le \text{CTH} \le 17 \text{ km}$ or (bottom) inner-core clouds with $\text{COD} \ge 100$, over (left) the whole lifetime, (middle) pre-lifetime maximum intensity (LMI), and (right) post-LMI period from July 4, 2002 to December 31, 2021. The shaded area is the two-sided 95% confidence interval of the best-fit lines with Spearman coefficient of correlation *r* and *p* value *P*. The data is binned into the mean of 10 deciles (i.e., every 10%) of the CF.

the same period (Hall et al., 2021). This lack of clear intensity changes with a declining rain rate suggests that there may have been an increase in the efficiency of the TC spin-up. There has been less column latent heat (i.e., rain) generated within the inner core during intensification but a similar ultimate LMI has been achieved. To diagnose this, we calculate a conversion ratio, C (Equation 1), as the ratio of the LMI kinetic energy to the total



Figure 4. Same as Figure 2 but for the annual-mean tropical cyclone (TC) inner-core rain rate between January 1, 1998 and December 31, 2019 for CAT 3–5 TCs.

Table 1

Linear Trend (% per Decade) of Lifetime-Mean Inner-Core Cloud Fraction (CF) of $15 \le CTH \le 17$ km, Lifetime-Mean Inner-Core Rain Rate, LMI and the Conversion Ratio C (Equation 1) for CAT 3–5 TCs Over Different Regions During 2000–2019 (2002–2021 for CF)

Region	CF (CTH)	Rain rate	LMI	С
Global	-11.5 ± 5.2	-13.6 ± 1.8	$+1.4 \pm 1.1^{*}$	$+20.7 \pm 4.0$
Western North Pacific	$-10.5 \pm 8.4*$	-10.7 ± 2.1	$+2.9 \pm 1.9*$	$+18.4 \pm 4.2$
Eastern North Pacific	-22.1 ± 9.7	-20.7 ± 6.2	$-10.3 \pm 6.9*$	$+43.3 \pm 11.2$
North Atlantic	$-17.7 \pm 14.0^{*}$	-20.5 ± 6.2	$-12.3 \pm 7.1^{*}$	$-1.3 \pm 9.9*$

Note. The asterisks denote insignificance (P > 0.05).

column latent heat for every major TC. This non-dimensional ratio has increased by about 21% per decade globally and in several basins (Table 1 and Figure S6 in Supporting Information S1).

4. Discussion and Conclusion

By examining MODIS/Aqua CLDPROP data, we find a decline of TC CF of inner-core clouds with high CTH or COD. Both the CF of inner-core clouds with high CTH ($15 \le z \le 17$ km) and that of inner-core very thick clouds with COD \geq 100 have significantly declined globally by about 10% per decade. The corresponding changes in outer-region CF are much smaller. The height interval $15 \le z \le 17$ km covers the usual CTHs of the TC eyewall (e.g., Duran & Molinari, 2018; Emanuel, 2018; Houze et al., 2009). We also show that thick clouds $(COD \ge 23.0)$ clearly dominate the inner core while the CF of the very thick clouds is the largest and much larger than that of optically thinner clouds. Therefore, these declining trends indicate that the inner-core convection of TCs has become shallower. Our TC trend is different from other reported cloud height trends (i.e., for all weather systems). Davies et al. (2017) showed that the global cloud height in MISR did not have a significant trend during 2000–2015. For the tropics, Richardson et al. (2022) found that the height of the tropical upper-tropospheric clouds (regardless of thickness) as measured by MODIS was increasing from 2002 to 2021. Hence, the declining trend of TC inner-core CTH implied in this study is quite anomalous and remarkable. It is worthwhile to note that the pattern shift (less higher/thicker clouds with more lower/thinner clouds) shown in Figure 1 and Figure S2 in Supporting Information S1 supports the case for a shallowing of TC inner core, but not the scenario of an expansion of the TC eye (or cloud-free area) in which case we would expect a CF decline across all CTH/COD or a decline in CF of high CTH/COD without an increase in CF of low CTH/COD.

We show very strong ranked correlations between TC inner-core rain rate and CF of cloud with high CTH or COD. This is a novel but expected result based on our understanding of rain and deep convection. Based on the correlation, we also re-examined the trend of TC rain rate in the same period. We confirm and extend past studies of a declining inner-core rain rate by considering a longer-time period, showing that the decrease is much more prominent during the intensification stage and noting that this decline has occurred while the LMI has been stable suggesting increased spin-up efficiency. Since the maximum potential intensity (MPI) of a TC is sensitive to the warm core height (Holland, 1997) and the actual intensity is correlated with the MPI (Wing et al., 2007), the stable LMI indicates a stable warm core height which seems contradictory to TC shallowing. However, the average height of warm core is $p \approx 250$ hPa (Holland, 1997; Ito & Yamamoto, 2022) or $z \approx 10$ km (Munsell et al., 2018; Stern & Zhang, 2016) which is substantially lower than the range ($15 \le z \le 17$ km) of the decrease in inner-core CF. Hence, it is possible that the warm core height, the MPI and LMI were stable during this period.

We study the inner-core and outer-region trends of TC rain rate from 1998 to 2019 which is a bit longer than done previously by Lavender and McBride (2021) (1998–2014), Tu et al. (2021) (1999–2018), and Guzman and Jiang (2021) (1998–2016). We confirm that the lifetime mean TC inner-core rain rate has declined. Both the observed weaker inner-core CF of high deep cloud and rain rate individually suggest weaker inner-core convection. It is remarkable that the percentage changes of TC inner-core rain rate and TC inner-core CF of cloud reaching $15 \le z \le 17$ km were similar. These two trends observed independently using different instruments, satellite platforms and variables but of the same convective process appear quantitatively consistent. The consistency between them reinforces the credibility of the individual satellite trends. Finally, we find similar granularity in the trends of CF and rain rate in terms of the TC life cycle. The intensification period shows the clearest trends. All of the pre-LMI declining trends of inner-core rain rate, CF of clouds with high CTH and CF of clouds with high COD were larger than the corresponding post-LMI trends. This shows further consistency between the observations indicating that the pre-LMI weakening of inner-core deep convection was more prominent. The absolute depth of convection is not straightforwardly or universally defined. The storm height can also be defined by a reflectivity threshold which captures denser clouds and the CTH as measured by brightness temperature will typically be larger. However, both are expected to increase with more intense rain (Song et al., 2020).

There is a suggestion that during the period examined, there has been an increase in the efficiency of the TC spin-up. We find that considerably less column latent heat by inner-core rain is generated during the spin-up to achieve similar intensities for major TCs. In a future study, a deeper understanding could be achieved by examining changes in the vertical profile of the inner-core heating rate which is important for the TC dynamics.

We also reveal that the average TC lifetime-mean intensity was declining during this period, with a more significant trend for the major TCs (Figure S8 in Supporting Information S1). This would be consistent with the shalowing TC inner-core convection and weakening inner-core rain rate. Additionally, the relatively stable LMI of TCs and the lower lifetime-mean intensity together imply a higher efficiency and a faster TC spin-up as reported by S. Wang et al. (2020).

What are the implications of these findings under future global warming? CTH can be partly understood through convective proxies such as convective available potential energy (CAPE) (e.g., Sherwood et al., 2004). There is an expectation of increases of convective proxies such as CAPE generally (Lepore et al., 2021) and by extension more high thick clouds. TC intensity and rain rate are also projected to increase (Knutson et al., 2020). However, this study shows that the TC inner-core CF of clouds with high CTH or COD and inner-core rain rate have both been decreasing for an extended period of time. This confirms that the vertical cloud structure and rain do change in tandem over decades. It does not negate the expectation of increased rain and larger CTHs in a future warmer world but shows that even a 20-year-long trend can oppose a longer-term projected change. Additionally, the declines of inner-core high thick cloud and rain rate have occurred while the intensity was stable suggesting an increased efficiency of TC spin-up.

Data Availability Statement

Best track data of TCs are provided by IBTrACS version 4 (the subset "since1980"), which is freely available at https://doi.org/10.25921/82ty-9e16. The data of cloud top height and cloud optical depth are obtained from MODIS/Aqua Level-3 Continuity Daily Cloud Properties Product version 1.1 (https://ladsweb.modaps.eosdis.nasa.gov/archive/allData/5111/CLDPROP_D3_MODIS_Aqua/; registration required for access). TC rainfall data are extracted from TRMM 3B42 version 7 (https://disc2.gesdisc.eosdis.nasa.gov/data/TRMM_L3/TRMM_3B42.7; registration required for access).

Acknowledgments

The research was supported with a generous donation by Dr. David Kinniburgh, the UK Centre for Greening of Finance and Investment (UKRI-NE/V017756/1) and the Singapore Green Finance Centre. The authors thank the two anonymous reviewers for their constructive comments which help improve this manuscript.

References

Adler, R. F., & Mack, R. A. (1984). Thunderstorm cloud height–rainfall rate relations for use with satellite rainfall estimation techniques. Journal of Applied Meteorology and Climatology, 23(2), 280–296. https://doi.org/10.1175/1520-0450(1984)023<0280:TCHRRF>2.0.CO;2

Chavas, D. R., Lin, N., Dong, W., & Lin, Y. (2016). Observed tropical cyclone size revisited. *Journal of Climate*, 29(8), 2923–2939. https://doi.org/10.1175/JCLI-D-15-0731.1

Davies, R., Jovanovic, V. M., & Moroney, C. M. (2017). Cloud heights measured by MISR from 2000 to 2015. *Journal of Geophysical Research:* Atmospheres, 122(7), 3975–3986. https://doi.org/10.1002/2017JD026456

Duran, P., & Molinari, J. (2018). Dramatic inner-core tropopause variability during the rapid intensification of Hurricane Patricia (2015). Monthly Weather Review, 146(1), 119–134. https://doi.org/10.1175/MWR-D-17-0218.1

Emanuel, K. A. (1987). The dependence of hurricane intensity on climate. *Nature*, 326(6112), 483–485. https://doi.org/10.1038/326483a0
Emanuel, K. A. (2005). Increasing destructiveness of tropical cyclones over the past 30 years. *Nature*, 436(7051), 686–688. https://doi.org/10.1038/nature03906

Emanuel, K. A. (2018). 100 years of progress in tropical cyclone research. *Meteorological Monographs*, 59, 15.1–15.68. https://doi.org/10.1 175/AMSMONOGRAPHS-D-18-0016.1

Guzman, O., & Jiang, H. (2021). Global increase in tropical cyclone rain rate. *Nature Communications*, 12(1), 5344. https://doi.org/10.1038/ s41467-021-25685-2

Hall, T. M., Kossin, J. P., Thompson, T., & McMahon, J. (2021). U.S. tropical cyclone activity in the 2030s based on projected changes in tropical sea surface temperature. *Journal of Climate*, 34(4), 1321–1335. https://doi.org/10.1175/JCLI-D-20-0342.1

Hendricks, E. A., Montgomery, M. T., & Davis, C. A. (2004). The role of "vortical" hot towers in the formation of tropical cyclone Diana (1984). Journal of the Atmospheric Sciences, 61(11), 1209–1232. https://doi.org/10.1175/1520-0469(2004)061<1209:TROVHT>2.0.CO;2

- Hill, K. A., & Lackmann, G. M. (2011). The impact of future climate change on TC intensity and structure: A downscaling approach. *Journal of Climate*, 24(17), 4644–4661. https://doi.org/10.1175/2011JCLI3761.1
- Holland, G. J. (1997). The maximum potential intensity of tropical cyclones. Journal of the Atmospheric Sciences, 54(21), 2519–2541. https:// doi.org/10.1175/1520-0469(1997)054<2519:tmpiot>2.0.co;2
- Houze, R. A., Jr., Lee, W.-C., & Bell, M. M. (2009). Convective contribution to the genesis of Hurricane Ophelia (2005). Monthly Weather Review, 137(9), 2778–2800. https://doi.org/10.1175/2009MWR2727.1
- Huffman, G. J. (2021). Expert developer guidance for TRMM Multi-satellite precipitation analysis (TMPA). Climate Data Guide, National Center for Atmospheric Research. Retrieved from https://climatedataguide.ucar.edu/climate-data/trmm-tropical-rainfall-measuring-mission
- Huffman, G. J., Bolvin, D. T., Nelkin, E. J., Wolff, D. B., Adler, R. F., Gu, G., et al. (2007). The TRMM Multi-satellite Precipitation Analysis (TMPA): Quasi-global, multiyear, combined-sensor precipitation estimates at fine scales. *Journal of Hydrometeorology*, 8(1), 38–55. https://doi.org/10. 1175/JHM560.1
- Ito, K., & Yamamoto, R. (2022). Thermodynamic and kinematic structure of tropical cyclones in the western North Pacific based on ACARS/ AMDAR. Frontiers in Earth Science, 10, 1058262. https://doi.org/10.3389/feart.2022.1058262
- Jiang, H. (2012). The relationship between tropical cyclone intensity change and the strength of inner-core convection. *Monthly Weather Review*, 140(4), 1164–1176. https://doi.org/10.1175/MWR-D-11-00134.1
- Kelley, O. A., Stout, J., & Halverson, J. B. (2004). Tall precipitation cells in tropical cyclone eyewalls are associated with tropical cyclone intensification. *Geophysical Research Letters*, 31(24), L24112. https://doi.org/10.1029/2004GL021616
- Kimball, S. K., & Mulekar, M. S. (2004). A 15-Year climatology of North Atlantic tropical cyclones. Part I: Size parameters. *Journal of Climate*, 17(18), 3555–3575. https://doi.org/10.1175/1520-0442(2004)017<3555:AYCONA>2.0.CO;2
- King, M. D., Menzel, W. P., Kaufman, Y. J., Tanré, D., Gao, B. C., Platnick, S., et al. (2003). Cloud and aerosol properties, precipitable water, and profiles of temperature and humidity from MODIS. *IEEE Transactions on Geoscience and Remote Sensing*, 41(2), 442–458. https://doi. org/10.1109/TGRS.2002.808226
- Knaff, J. A., Longmore, S. P., & Molenar, D. A. (2014). An objective satellite-based tropical cyclone size climatology. *Journal of Climate*, 27(1), 455–476. https://doi.org/10.1175/JCLI-D-13-00096.1
- Knapp, K. R., Diamond, H. J., Kossin, J. P., Kruk, M. C., & Schreck, C. J. (2018). International Best Track Archive for Climate Stewardship (IBTrACS) project, version 4. [since 1980]. NOAA National Centers for Environmental Information. https://doi.org/10.25921/82ty-9e16
- Knapp, K. R., Kruk, M. C., Levinson, D. H., & Neumann, C. J. (2010). The International Best Track Archive for Climate Stewardship (IBTrACS): Unifying tropical cyclone best track data. *Bulletin of the American Meteorological Society*, 91(3), 363–376. https://doi.org/ 10.1175/2009BAMS2755.1
- Knutson, T. R., Camargo, S. J., Chan, J. C. L., Emanuel, K. A., Ho, C.-H., Kossin, J. P., et al. (2020). Tropical cyclones and climate change assessment. Part II: Projected response to anthropogenic warming. *Bulletin of the American Meteorological Society*, 101(3), E303–E322. https://doi.org/ 10.1175/BAMS-D-18-0194.1
- Knutson, T. R., Sirutis, J. J., Zhao, M., Tuleya, R. E., Bender, M., Vecchi, G. A., et al. (2015). Global projections of intense tropical cyclone activity for the late twenty-first century from dynamical downscaling of CMIP5/RCP4.5 scenarios. *Journal of Climate*, 28(18), 7203–7224. https://doi.org/10.1175/JCLI-D-15-0129.1
- Knutson, T. R., & Tuleya, R. E. (1999). Increased hurricane intensities with CO₂-induced warming as simulated using the GFDL hurricane prediction system. *Climate Dynamics*, 15(7), 503–519. https://doi.org/10.1007/s003820050296
- Kossin, J. P., Emanuel, K. A., & Vecchi, G. A. (2014). The poleward migration of the location of tropical cyclone maximum intensity. *Nature*, 509(7500), 349–352. https://doi.org/10.1038/nature13278
- Lavender, S. L., & McBride, J. L. (2021). Global climatology of rainfall rates and lifetime accumulated rainfall in tropical cyclones: Influence of cyclone basin, cyclone intensity and cyclone size. *International Journal of Climatology*, 41(S1), E1217–E1235. https://doi.org/10.1002/ joc.6763
- Lepore, C., Abernathey, R., Henderson, N., Allen, J. T., & Tippett, M. K. (2021). Future global convective environments in CMIP6 models. *Earth's Future*, 9(12), e2021EF002277. https://doi.org/10.1029/2021EF002277
- Lonfat, M., Marks, F. D., & Chen, S. (2004). Precipitation distribution in tropical cyclones using the Tropical Rainfall Measuring Mission (TRMM) microwave imager: A global perspective. *Monthly Weather Review*, 132(7), 1645–1660. https://doi.org/10.1175/1520-0493(2004)132<1645:pditcu>2.0.co;2
- Marchand, R. (2013). Trends in ISCCP, MISR, and MODIS cloud-top-height and optical-depth histograms. Journal of Geophysical Research: Atmospheres, 118(4), 1941–1949. https://doi.org/10.1002/jgrd.50207
- Marchand, R., Ackerman, T. A., Smyth, M., & Rossow, W. B. (2010). A review of cloud top height and optical depth histograms from MISR, ISCCP, and MODIS. Journal of Geophysical Research: Atmospheres, 115(D16), D16206. https://doi.org/10.1029/2009JD013423
- Munsell, E. B., Zhang, F., Braun, S. A., Sippel, J. A., & Didlake, A. C., Jr. (2018). The inner-core temperature structure of Hurricane Edouard (2014): Observations and ensemble variability. *Monthly Weather Review*, 146(1), 135–155. https://doi.org/10.1175/MWR-D-17-0095.1
- Platnick, S., Meyer, K. G., Hubanks, P., Holz, R., Ackerman, S. A., & Heidinger, A. K. (2019). VIIRS Atmosphere L3 Cloud Properties Product. Version-1.1. (NASA Level-1 and Atmosphere Archive & Distribution System (LAADS) Distributed Active Archive Center (DAAC). Goddard Space Flight Center. https://doi.org/10.5067/VIIRS/CLDPROP_D3_VIIRS_SNPP.011
- Richardson, M. T., Roy, R. J., & Lebsock, M. D. (2022). Satellites suggest rising tropical high cloud altitude: 2002–2021. Geophysical Research Letters, 49(10), e2022GL098160. https://doi.org/10.1029/2022GL0981601
- Schenkel, B. A., Lin, N., Chavas, D. R., Vecchi, G. A., Oppenheimer, M., & Brammer, A. (2018). Lifetime evolution of outer tropical cyclone size and structures as diagnosed from reanalysis and climate model data. *Journal of Climate*, 31(19), 7985–8004. https://doi.org/10.1175/ JCLI-D-17-0630.1
- Schubert, W. H., & McNoldy, B. D. (2010). Application of the concepts of Rossby Length and Rossby Depth to tropical cyclone dynamics. Journal of Advances in Modeling Earth Systems, 2, 7. https://doi.org/10.3894/JAMES.2010.2.7
- Sherwood, S. C., Minnis, P., & McGill, M. (2004). Deep convective cloud-top heights and their thermodynamic control during CRYSTAL-FACE. *Journal of Geophysical Research: Atmospheres, 109*(D20), D20119. https://doi.org/10.1029/2004JD004811
- Simpson, J., Halverson, J. B., Ferrier, B. S., Petersen, W. A., Simpson, R. H., Blakeslee, R., & Durden, S. L. (1998). On the role of "hot towers" in tropical cyclone formation. *Meteorology and Atmospheric Physics*, 67(1–4), 15–35. https://doi.org/10.1007/BF01277500
- Song, H.-J., Kim, S., Roh, S., & Lee, H. (2020). Difference between cloud top height and storm height for heavy rainfall using TRMM measurements. Journal of the Meteorological Society of Japan, 98(5), 901–914. https://doi.org/10.2151/jmsj.2020-044
- So, D., & Shin, D.-B. (2018). Classification of precipitating clouds using satellite infrared observations and its implications for rainfall estimation. Quarterly Journal of the Royal Meteorological Society, 144(S1), 133–144. https://doi.org/10.1002/qj.3288



- Stern, D. P., & Zhang, F. (2016). The warm-core structure of Hurricane Earl (2010). Journal of the Atmospheric Sciences, 73(8), 3305–3328. https://doi.org/10.1175/JAS-D-15-0328.1
- Tapiador, F. J., Marcos, C., & Sancho, J. M. (2019). The convective rainfall rate from cloud physical properties algorithm for Meteosat second-generation satellites: Microphysical basis and intercomparisons using an object-based method. *Remote Sensing*, 11(5), 527. https:// doi.org/10.3390/rs11050527
- Tropical Rainfall Measuring Mission (TRMM). (2011). TRMM (TMPA) rainfall estimate L3 3 hour 0.25 degree x 0.25 degree V7. (NASA Goddard Earth Sciences Data and Information Services Center (GES DISC). https://doi.org/10.5067/TRMM/TMPA/3H/7
- Tu, S., Xu, J., Chan, J. C. L., Huang, K., Xu, F., & Chiu, L. S. (2021). Recent global decrease in the inner-core rain rate of tropical cyclones. *Nature Communications*, 12(1), 1948. https://doi.org/10.1038/s41467-021-22304-y
- Wang, S., Rashid, T., Throp, H., & Toumi, R. (2020). A shortening of the life cycle of major tropical cyclones. *Geophysical Research Letters*, 47(14), e2020GL088589. https://doi.org/10.1029/2020GL088589
- Wang, S., & Toumi, R. (2016). On the relationship between hurricane cost and the integrated wind profile. *Environmental Research Letters*, 11, 114005. https://doi.org/10.1088/1748-9326/11/11/114005
- Wang, S., & Toumi, R. (2018). A historical analysis of the mature stage of tropical cyclones. International Journal of Climatology, 38(5), 2490–2505. https://doi.org/10.1002/joc.5374
- Wang, S., & Toumi, R. (2021a). Recent migration of tropical cyclones toward coasts. Science, 371(6528), 514–517. https://doi.org/10.1126/ science.abb9038
- Wang, S., & Toumi, R. (2021b). Recent tropical cyclone changes inferred from ocean surface temperature cold wakes. *Scientific Reports*, 11(1), 22269. https://doi.org/10.1038/s41598-021-01612-9
- Wang, S., & Toumi, R. (2022). More tropical cyclones are striking coasts with major intensities at landfall. *Scientific Reports*, 12(1), 5236. https://doi.org/10.1038/s41598-022-09287-6
- Wang, Z., Montgomery, M. T., & Fritz, C. (2009). A first look at the structure of the wave pouch during the 2009 PREDICT–GRIP dry runs over the Atlantic. *Monthly Weather Review*, 140(4), 1144–1163. https://doi.org/10.1175/MWR-D-10-05063.1
- Wing, A. A., Sobel, A. H., & Camargo, S. J. (2007). Relationship between the potential and actual intensities of tropical cyclones on interannual time scales. *Geophysical Research Letters*, 34(8), L08810. https://doi.org/10.1029/2006GL028581
- Yamada, Y., Oouchi, K., Satoh, M., Tomita, H., & Yanase, W. (2010). Projection of changes in tropical cyclone activity and cloud height due to greenhouse warming: Global cloud-system-resolving approach. *Geophysical Research Letters*, 37(7), L07709. https://doi.org/10.1029/2010 GL042518
- Yanai, M., Esbensen, S., & Chu, J.-H. (1973). Determination of bulk properties of tropical cloud clusters from large-scale heat and moisture budgets. *Journal of the Atmospheric Sciences*, 30(4), 611–627. https://doi.org/10.1175/1520-0469(1973)030<0611:DOBPOT>2.0.CO;2
- Zhai, A. R., & Jiang, J. H. (2014). Dependence of US hurricane economic loss on maximum wind speed and storm size. *Environmental Research* Letters, 9(6), 064019. https://doi.org/10.1088/1748-9326/9/6/064019